Author's response to reviews

Title: Proteinuria and hematuria are associated with acute kidney injury and mortality in critically ill patients: a retrospective observational study

Authors:

Seung Seok Han (hansway@medimail.co.kr)
Shin Young Ahn (ahnshinyoung712@gmail.com)
Jiwon Ryu (bboddo5@hanmail.net)
Seon Ha Baek (haya2001@hanmail.net)
Ho Jun Chin (mednep@snubh.org)
Ki Young Na (kyna@snubh.org)
Dong-Wan Chae (cdw1302@snubh.org)
Sejoong Kim (imsejoong@hanmail.net)

Version: 4 Date: 22 May 2014

Author's response to reviews: see over
Editorial Requests

Thank you for reuploading your manuscript. Unfortunately the title page is still missing the email addresses of your co-authors. Once these have been included we will be able to proceed with acceptance.

We missed the description of e-mails in previous version of manuscript. We have included e-mail addresses of all authors in the revised version of manuscript.